Author's response to reviews

Title: Risk factors of diarrheal disease in under-five children among health extension model and non-model families in Sheko district rural community, Southwest Ethiopia

Authors:

Teklemichael Gebru (teklemichaelgebru@gmail.com)
Mohammed Taha (tahamohammed2009@gmail.com)
Wondwosen Kassahun (wondkyt@yahoo.com)

Version: 7 Date: 14 April 2014

Author's response to reviews: see over
Cover letter
Date: April/2014
To,
BMC reviews
Title: Risk factors of diarrheal disease in under-five children among health extension model and non-model families in Sheko district rural community, Southwest Ethiopia: Comparative cross-sectional study
Authors:
   Teklemichael Gebru (teklemichaelgebru@gmail.com)
   Mohammed Taha (tahamohammed2009@gmail.com)
   Wondwosen Kassahun (wondkyt@yahoo.com)
Author's response to reviews: see over

We were pleased to have an opportunity to revise our paper and we have greatly appreciated the reviewers’ comments and suggestions were very helpful overall. In revising the paper, we have carefully considered reviewers’ comments and suggestions on our revised submission. As instructed, we have attempted to succinctly explain changes made in reaction to all comments and we hope you agree. After providing a brief overview of ways in which the paper was revised, we reply to each comment in point-by-point fashion as follows:

Referee: Aderaw Anteneh
Major Compulsory Revisions
Comment 1: “In the current version of the paper the authors tried to address the comments given satisfactorily. But I have a comment on the references, some of the references are not written properly. Please check again how references are written in BMC journal. For example, reference 7 should not be written like that the correct one is: CSA and ORC Macro (2006): Ethiopian Demographic and Health Survey 2005, Addis Ababa, Ethiopia and Calverton, Maryland, USA. There are also many references in your document that are written wrongly.”
Response 1: Previously we use Zotero to organise our reference but now as suggested, your constrictive comment we were tried to correct in the revised document.

Thank you, we hope you now find the manuscript acceptable for publication in BMC public health!

Sincerely yours,
Teklemichael Gebru